



# Article Performance Evaluation of Retail Warehouses: A Combined MCDM Approach Using G-BWM and RATMI

Abdullah M. Barasin<sup>1</sup>, Ammar Y. Alqahtani<sup>1,\*</sup> and Anas A. Makki<sup>2</sup>

- <sup>1</sup> Department of Industrial Engineering, Faculty of Engineering, King Abdulaziz University, P.O. Box 80204, Jeddah 21589, Saudi Arabia; aahmadbarasin@stu.kau.edu.sa
- <sup>2</sup> Department of Industrial Engineering, Faculty of Engineering—Rabigh, King Abdulaziz University, P.O. Box 80204, Jeddah 21589, Saudi Arabia; nhmakki@kau.edu.sa
- \* Correspondence: aaylqahtani@kau.edu.sa

Abstract: Background: The retail sector has experienced significant growth in recent years, necessitating efficient supply chain management and sustainable logistics practices. Evaluating the performance of retail warehouses is crucial for meeting customer expectations and enhancing operational efficiency. Methods: This study employed a combined multi-criteria decision-making (MCDM) approach, using the group best-worst method (G-BWM) for weighting criteria and ranking the alternatives based on the trace-to-median index (RATMI) for warehouse ranking. The performance criteria were cost, quality, time, productivity, and safety. Data were collected from four mega retail warehouses in the western region of Saudi Arabia for evaluation and analysis. Results: The evaluation of retail warehouse performance using the MCDM approach provided valuable insights for decision-makers and warehouse experts. The criteria weights were determined using the G-BWM, and the RATMI enabled the ranking of the warehouses based on their weighted performance scores. The results highlight the strengths and weaknesses of each warehouse, facilitating strategic planning, resource allocation, and operational improvements. Conclusions: This study presents a novel combined MCDM performance evaluation approach for retail warehouses. The study has implications for effective decision-making processes, resource allocation, and operational efficiency. Furthermore, it serves as a foundation for future research, exploring additional dimensions of warehouse performance and enabling sustainable logistics within the broader supply chain context.

Keywords: retail; warehouses; performance; evaluation; MCDM; G-BWM; RATMI

# 1. Introduction

A warehouse is one of the most critical parts of many companies, essential to facilitating trade. With a robust warehouse management system, a business can satisfy customer demand. Additionally, it helps guarantee that the products are affordable, easily accessible, and delivered quickly to a network of customers. However, if not properly structured and managed, it may prevent a business from competing effectively, locally and worldwide. One of the most crucial ways to enhance warehouses and assist managers in continuously monitoring their operations is warehouse performance measurement. Management must establish a variety of criteria to gauge warehouse performance. Based on these criteria, they can determine whether the warehouse is performing well.

Each retail warehouse in Saudi Arabia's western region has a set of criteria for measuring its performance. Warehouses occupy a significant position in the economy of Saudi Arabia. They are among the most significant national focuses of development and improvement to increase the efficiency of the country's economy in logistics [1]. With Vision 2030, the government began to improve Saudi Arabia's logistics infrastructure, and as it grows, so does warehousing. The warehouse market expanded by 2.8% between 2015 and 2019, and it is expected to grow more in the future.



Citation: Barasin, A.M.; Alqahtani, A.Y.; Makki, A.A. Performance Evaluation of Retail Warehouses: A Combined MCDM Approach Using G-BWM and RATMI. *Logistics* **2024**, *8*, 10. https://doi.org/10.3390/ logistics8010010

Academic Editors: Mladen Krstić, Željko Stević and Snežana Tadić

Received: 12 December 2023 Revised: 7 January 2024 Accepted: 10 January 2024 Published: 12 January 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). According to Statista, the operating revenue of warehousing was USD 8.84 billion between 2010 and 2017. The value of e-commerce warehousing in the Middle East could reach USD 500 million by 2024.

Because of the COVID-19 pandemic, the world has encountered numerous warehousing issues, as has Saudi Arabia. The pandemic has significantly impacted logistics, and Saudi Arabia must overcome some challenges to continue warehouse expansion.

According to Colliers International, the average price of warehouse space in Jeddah, Saudi Arabia, was SAR 131 (USD 35) per square meter in the first quarter of 2020. This high price means that retail warehouses must reduce operational expenses, optimize their management systems, and solve problems with cutting-edge scientific techniques.

Most real-world decision-making challenges require the simultaneous consideration of several competing criteria and objectives. Similar challenges occur in various professions, including engineering, medicine, and business. Multi-criteria decision-making (MCDM) is concerned with structuring and resolving problems involving multiple criteria and conflicting goals. With the increase in the number of warehouses in Saudi Arabia, the competition between them has increased, and every warehouse has evaluation criteria to prove its superiority over others. Therefore, this paper explains how combined approaches between the group best–worst method (G-BWM) and ranking the alternatives based on the trace-to-median index (RATMI) will help decision-makers evaluate and rank selected Saudi Arabian warehouse alternatives from best to worst. Both quantitative and qualitative factors can be considered in this MCDM problem.

The main objective of this study was to evaluate and compare the performance of retail warehouses in the western region of Saudi Arabia to identify the best-performing warehouse. An MCDM approach was used to systematically assess the warehouses against key performance criteria related to time, cost, quality, productivity, and safety. The G-BWM method was employed to determine the weights of the various criteria. The RATMI technique was used to rank the warehouses from best to worst based on their weighted performance scores. The results of this evaluation may provide decision-makers and warehouse managers with valuable insights into the strengths and weaknesses across the warehouses. This will allow for more effective strategic planning, resource allocation, and operational improvements to maximize the overall warehouse efficiency and competitiveness in the Saudi retail sector.

This study aimed to fill this research gap by presenting a novel approach to evaluating retail warehouse performance. Previous studies have used various methodologies, but a comprehensive and integrated method that considers multiple criteria and provides a robust assessment is needed. Furthermore, existing approaches often lack a systematic ranking mechanism to differentiate the performance of retail warehouses.

This research introduces a combined MCDM approach to address these limitations, specifically employing the G-BWM for criteria weighting and the RATMI for warehouse ranking. This innovative approach allows for a more holistic evaluation of retail warehouses by considering key performance criteria such as cost, quality, time, productivity, and safety. In addition to the novel methodology, this study distinguishes itself by using a unique dataset collected from four mega retail warehouses in the western region of Saudi Arabia. This dataset provides valuable insights into the performance of retail warehouses in a specific geographic context, enabling a more focused analysis and relevant findings. The implications of this research are significant. By employing the G-BWM and RATMI techniques, decision-makers and warehouse managers can gain valuable insights into the strengths and weaknesses of retail warehouses, facilitating strategic planning, resource allocation, and operational improvements. This approach contributes to more effective decision-making processes and enhances the overall operational efficiency of retail warehouses.

Moreover, this study opens avenues for future research endeavors. Exploring additional dimensions of warehouse performance and their influence on sustainable logistics represents a promising area for further investigation. Expanding the understanding of retail warehouse performance means that researchers and practitioners can work toward achieving more sustainable and efficient supply chains.

The rest of this paper is structured as follows. Section 2 provides a literature review on key topics related to warehouse performance measurement, applications of MCDM tools in warehousing, and commonly used MCDM techniques. Section 3 outlines the study methodology, including the identification of warehouses and criteria, determination of criteria weights, and evaluation of warehouse performance as alternatives. Section 4 presents the results, and Section 5 discusses the findings and results of the analysis. Finally, Section 6 summarizes the main conclusions and limitations of the study, as well as recommendations for future work.

# 2. Literature Review

This literature review provides an overview of previous research relevant to the topics studied in this paper. It is divided into three subsections to comprehensively cover the key areas. The first reviews studies on warehouse performance indicators and the key criteria considered essential for measuring warehouse operations. Several classification frameworks and findings from major literature reviews in this domain are summarized. The second subsection examines the available literature applying MCDM tools to assess warehouse performance evaluation problems. It outlines some notable applications and methodologies used in related contexts such as warehouse design, layout, and location selection. Finally, the third subsection explores common MCDM techniques discussed in the literature. An overview of various MCDM approaches is presented, highlighting relevant methods such as Elimination Et Choix Traduisant la Realité (ELECTRE), simple additive weighting (SAW), analytic hierarchy analysis (AHP), the best–worst method (BWM), and VlseKriterijuska Optimizacija I Komoromisno Resenje (VIKOR) [2]. This section thus provides the relevant background and establishes the need for the combined methodology evaluated in this study.

# 2.1. Warehouse Key Performance Indicators

Rouwenhorst et al. [3] classified warehouses from the perspectives of processes, resources, and organizational structures. They conducted a thorough literature review, concluding that most studies focused primarily on analysis and not warehouse design. Gu et al. [4] extensively reviewed warehouse operation planning problems. The problems were classified according to basic warehouse functions: receiving, storage, order picking, and shipping. The literature in each category was summarized, emphasizing the characteristics of various decision-support models and solution algorithms. Storage and order-picking functions impacted the warehouse operational performance the most. Staudt et al. [5] emphasized measuring operational warehouse performance. Performance indicators were extracted from relevant papers using the content analysis method and were categorized based on time, cost, quality, and productivity dimensions. Vaturalae et al. [6] discussed the Malaysian hypermarket retail sector in detail, followed by a thorough literature review on the warehouse management system. The literature review focused on the hypermarket retailers' warehouses, which are essential in the supply chain, facilitating the movement of materials between the supplier and the customer. A model presented by Ramirez-Malule et al. [7] identified variables that significantly impact warehouse performance with picker-to-parts storage systems, considering the dynamic nature of the model's processes and the possibility of non-linear relationships among its variables, as well as the simultaneous occurrence of seasonal demand and long and short product life-cycles. Phyllis [8] applied warehouse performance measurement in the case of a medium-sized warehouse in Nakuru Town. The warehouse performance indicators were classified into five categories: productivity, utilization, quality, time, and financial. The findings indicated that the most critical performance indicators were productivity for receiving, space utilization for storage, cycle time for order picking, and productivity for shipping. A study on measuring warehouse performance in third-party logistics (3PL) service providers was

conducted by Ghaouta and Okar [9]. Their research had three main objectives: to review warehouse key performance indicators (KPIs) and categories using the systematic literature review method, to investigate categories and determine their relative importance using the Q-sort method, and to validate the order of performance measure categories using a single case study. The study grouped 30 KPIs into five categories and four subcategories using an integrated research methodology.

# 2.2. Warehouse Performance Evaluation Using MCDM Tools

Few studies have offered information about the MCDM tools employed in the warehouse performance evaluation context. A study by Kusrini et al. [10] assessed the performance of retail warehouses in supermarkets in Central Java and Yogyakarta, Indonesia. The criteria were weighted using the AHP approach. After assessing the warehouse's performance, the final score was calculated using the SNORM method. MCDM tools have a wide range of applications in warehouse location problems as well as issues with warehouse architecture and design. Al Amin et al. [11] employed AHP and the technique for order of preference by similarity to ideal solution (TOPSIS) to select the best warehouse among five based on five specified criteria (unit price, stock holding capacity, average distance to factory, flexibility, and layout). Demircioğlu and Ozceylan [12] conducted a thorough literature review using pertinent keywords in several worldwide databases to explore MCDM applications in warehouse layout and design. AHP, ELECTRE, and the Preference Ranking Organization Method for Enrichment Evaluation (PROMETHEE) were the most frequently used techniques, which are carried out by applying MCDM methods. To manage the benefits of employing many MCDM methods in a given field, MCDM approaches are employed in an integrated manner. Ulutas et al. [13] suggested an integrated grey MCDM model including the grey preference selection index (GPSI) and grey proximity indexed value (GPIV) to determine the most appropriate warehouse location for a supermarket. Twelve criteria were employed to compare the effectiveness of five potential warehouse locations. The optimum warehouse location was chosen using a combination of GPSI and GPIV algorithms [13]. Fuzzy extensions of MCDM techniques have also been used in other contexts [14–16]. In the context of warehouses, Bairagi [17] employed fuzzy MCDM to assess the location. The warehouse selection indicator, known as the benefit-cost ratio (BCR) [18], is assessed using the aggregate modified weighted value of the warehouse locations' normalized scores.

### 2.3. MCDM Tools

MCDM tools are used to deal with challenging real-world problems since they can assess several options and choose the optimal one [19]. The literature includes many different MCDM approaches. The ELECTRE tool allows for identifying and eliminating options that are outranked by others, leaving a set of appropriate actions [20]. ELECTRE is a method that experts use to assess the effects of criteria and contrast them with one another based on the anticipated performance [21]. Another MCDM technique, SAW, aims to assess the effectiveness of various solutions [22,23]. The basic concept of SAW is to determine the weighted total of performance ratings for each alternative across all criteria. MCDM also includes stepwise weight assessment ratio analysis (SWARA). When using the SWARA technique, the criteria needs are ranked according to their importance by a group of experts [24]. Saaty [25] proposed the AHP method in the 1980s. AHP provides a logical framework for a decision that must be made by weighing the requirements and available alternatives and connecting those components to the primary objective. The BWM is an MCDM technique that Rezaei introduced in 2015 [26]. It can be used to evaluate alternative solutions in consideration of the criteria and assess the applicability of the criteria to discover a solution to accomplish the issue's main goal(s). Based on the BWM, a novel approach to group decision-making problems called the G-BWM was developed by Haseli et al. [27]. This approach assists in the analysis of decision-makers' preferences for employing the BWM structure for democratic decision-making. Another MCDM technique

is VIKOR. This technique involves weighing and choosing options based on competing criteria by outlining in detail how close each alternative is to the best hypothetical answer [28–31]. TOPSIS is another MCDM tool that has been used successfully in many applications [32–36]. Its fundamental objective is to find an optimal solution with the largest and lowest distances to the positive and negative ideal solutions, respectively [37]. The Saudi National Commission for Academic Accreditation and Evaluation (NCAAA) created the Self-Evaluation Scale to evaluate higher education programs. The present study used the TOPSIS technique to compare NCAAA's original performance criteria and the proposed evaluation sub-criteria [38]. Abdulaal and Bafail [39] developed two new approaches: ranking alternatives based on median similarity (RAMS) and RATMI. RAMS is a developed method that utilizes the ranking alternatives perimeter similarity (RAPS) [40]. The RATMI technique combines the RAMS method with the multiple criteria ranking by alternative trace methodology, using a majority index and the concept of the VIKOR method [41]. RATMI is a new technique that has been applied in recent studies [39,40,42,43].

The rationale for using the G-BWM for criteria weighting and the RATMI for warehouse ranking in evaluating warehouse performance was based on their respective advantages and suitability for addressing the research objectives. The G-BWM technique was chosen for criteria weighting due to its ability to capture the collective preferences of a group of decision-makers. In evaluating warehouse performance, multiple criteria contributing to operational effectiveness and customer satisfaction must be considered. However, assigning appropriate weights to these criteria can be challenging since decision-makers may have different perspectives and priorities. The G-BWM addresses this challenge by involving experts or decision-makers who participate in pairwise criteria comparisons. By identifying the best and worst criteria in each pairwise comparison, the G-BWM aggregates the individual preferences to derive consensus-based weights for the criteria. This group-based approach ensures that the evaluation framework considers diverse perspectives and avoids undue influence from a single decision-maker. The RATMI technique is employed for warehouse ranking to evaluate warehouse performance systematically and comprehensively. Traditional ranking methods often suffer from limitations such as subjectivity, inconsistency, and a lack of consideration for the interrelationships between criteria. The RATMI overcomes these limitations by using the concept of the median performance profile. It traces the performance of individual warehouses relative to the median performance profile, capturing each warehouse's relative strengths and weaknesses across multiple criteria. This approach offers a more objective and comparative assessment, enabling decision-makers to identify the top-performing and underperforming warehouses in the context of the entire dataset.

The evaluation framework ensures a comprehensive and robust assessment of warehouse performance by combining the G-BWM for criteria weighting and the RATMI for warehouse ranking. The G-BWM incorporates group preferences and diverse perspectives, and the RATMI provides an objective and comparative ranking mechanism. Together, these techniques enhance the evaluation framework's accuracy, reliability, and applicability, enabling decision-makers to make informed decisions and drive operational improvements in retail warehouses.

In this paper, two combined MCDM techniques, comprising the G-BWM technique, were used to group the criteria and give weights to each one. The RATMI technique ranks the alternatives based on the weight given by the G-BWM. The G-BWM allows for group decision-making in determining the criteria weights. Since the evaluations involve subjectivity from multiple warehouse experts, using their collective judgments through G-BWM provides a more democratic weighting process. The RATMI enables a comprehensive ranking of the warehouse alternatives based on the weighted performance scores across all criteria. This helps to systematically produce an overall performance evaluation and ranking. Combining both techniques allows them to complement each other. The G-BWM provides the weights as input for the RATMI, which then uses those weights to generate the final ranking. This makes the evaluation more robust by leveraging the strengths of both

methods. Both the G-BWM and the RATMI have been validated in previous studies for effectiveness in group MCDM problems. Applying them together extends their combined application to the warehouse performance evaluation context. The combination approach addresses the limitations of the individual techniques and provides the cross-validation of the results through the convergence of the two methods on a common solution or ranking. This enhances the reliability of findings. In summary, the researchers chose a combined G-BWM and RATMI approach to leverage their synergies, produce a rigorous yet democratic weighting process, and generate a validated overall performance ranking for strategic decision-making. This research aimed to contribute to the existing literature by filling the research gap and providing decision-makers and warehouse managers with a more robust and integrated method for evaluating retail warehouse performance. By addressing the need for a systematic ranking mechanism and considering multiple performance criteria, this study offers a novel perspective that enhances the understanding of warehouse effectiveness and supports data-driven decision-making in the retail sector.

# 3. Materials and Methods

The proposed framework of the methodology used for the performance evaluation and ranking of a group of chosen retail warehouses in Saudi Arabia's western region is shown in Figure 1. The methodology consists of three consecutive stages. The researchers conducted initial visits to various warehouses in the western region of Saudi Arabia in Stage 1. This allowed for the gaining of a first-hand understanding of warehouse operations, screening and selecting the four most suitable warehouses for the study based on predefined criteria, and building rapport with warehouse managers. Being on-site allowed for the introduction of the objectives of the performance evaluation study, distributing questionnaires to collect the required primary data, obtaining signed consent forms, and gathering any available secondary data on existing measurement processes. The visits proved crucial for gathering important operational context beyond documents, resolving ambiguities in the questionnaire, screening appropriate case warehouses, facilitating data collection processes, and strengthening the engagement of participants. Overall, conducting the preliminary warehouse visits was instrumental in setting the foundation for the subsequent stages of the MCDM evaluation approach. A questionnaire was designed and delivered to the warehouses and completed by warehouse experts to determine how they evaluated the performance of the warehouse's primary activities from their perspective. Each warehouse expert had different years of experience and education levels; Table A1 in Appendix A shows the warehouse experts' profiles. The G-BWM was employed in Stage 2 of the proposed methodology to calculate the criteria weights. In Stage 3, the chosen warehouses were ranked using RATMI. The details of each stage are provided in the following subsections.

# 3.1. Stage 1: Identifying the Warehouses and Their Performance Criteria

Step 1: This step deals with choosing the warehouses (alternatives). After visiting different warehouses in the western region of Saudi Arabia, four mega warehouses that have been operating for a long time in the retail sector were selected.

Step 2: In this step, a questionnaire is designed and distributed to the chosen warehouses to collect the required data for the second stage of the proposed methodology. The questionnaire contains 18 sections. Each section discusses different topics, starting from general information, respondents' profiles, the warehouses' primary activities, the KPIs of each primary activity, and the performance criteria of each KPI. The sections included 54 questions in total. For each question, the warehouse's experts were asked to rate on a scale from 1 to 9 to indicate their preference for the best activity over all the other activities and for all the other activities over the worst activity. Table 1 shows the scale meaning.

Step 3: This step shows the primary activities that all the chosen warehouses deal with.



Figure 1. The proposed framework for retail warehouses' performance evaluation.

Table 1.	Integer	scale	definition.
----------	---------	-------	-------------

Integer Scale	Definition
1	Equally important
2, 3, 4	Moderately important
5, 6	Strongly important
7,8	Very strongly important
9	Absolutely important

Step 4: This step is to find the KPIs for the primary activities. The KPIs of the chosen warehouses in this research study have been agreed to be the same, and are primarily used in different warehouses worldwide [3–9].

Step 5: Each KPI identified in Step 4 has indicators that give more accurate measurements of the performance of the warehouses.

#### 3.2. Stage 2: Determining the Criteria Weights

Step 6: This step will assign the weights to the warehouse performance criteria. The MCDM technique that will be used to give the weight for the criteria is G-BWM. The technique will be applied to calculate the weights three times: first on the primary warehouse activities, then on the KPIs of the primary activities, and last on the performance criterion (indicators). The G-BWM [27] technique has several steps to apply as follows:

Step 6.1: Considering a set of criteria  $\{C_1, C_2, ..., C_m\}$  to achieve a decision through the decision makers  $\{DM_1, DM_2, ..., DM_n\}$ , where *m* is a set of criteria and *n* is the number of decision-makers. Each decision maker (i.e., warehouse expert) generally selects the best and worst criteria.

Step 6.2: A pairwise comparison using crisp integer score values between 1 and 9 is performed. The numerical scales of 1 to 9 are used to determine the relative importance of the pairwise comparisons. Here,  $a_{ij} = 1$  represents the equal importance of the criterion *i* and over criterion *j*. Moreover,  $a_{ij} = 9$  signifies the extreme importance preference of criterion *i* over criterion *j*, as shown in Table 1.

The vector of the best criterion over other criteria would be

$$A_{Bj} = (a_{B1}, a_{B2}, \cdots, a_{Bm}), \ j = 1, 2, 3, \cdots, m$$
 (1)

 $A_{Bi}$  denotes the relative importance value of the best criterion over criterion *j*.

Step 6.3: The crisp numbers from 1 to 9 (Table 1) will also be used to perform the pairwise comparison in this step. As mentioned in the previous step,  $a_{ij} = 1$  represents the equal importance of criterion *i* and over criterion *j*. Moreover,  $a_{ij} = 9$  signifies the extreme importance preference of criterion *i* over criterion *j*. So, the vector of all criteria over the worst criterion would be

$$A_{jW} = (a_{1W}, a_{2W}, \cdots, a_{mW}), \ j = 1, 2, 3, \cdots, m$$
 (2)

 $A_{iW}$  stands for the relative importance value of criterion *j* over the worst criterion.

Step 6.4: The decision-makers who select the best and worst criteria belong to one group,  $G_i$ , where i = 1, 2, ..., k, and k is the number of groups.

Step 6.5: After creating the group, the result of grouping decision makers would be

$$Group DM = (G_1, G_2, \dots, G_k) \tag{3}$$

Step 6.6: In this step, the geometric mean in each group is used to evaluate the decision-makers for each Total  $A_{Bi}$  and Total  $A_{iW}$ . For each group  $(G_1, G_2, ..., G_k)$ :

$$\left(\prod_{i=1}^{n} \operatorname{Total} a_{Bj(DM_{i})}\right)^{\frac{1}{n}} = \sqrt[n]{a_{Bj(DM_{1})} \times a_{Bj(DM_{2})} \times \dots A_{Bj(DM_{n})}}$$
(4)

$$\left(\prod_{i=1}^{n} \operatorname{Total} a_{jW(DM_{i})}\right)^{\frac{1}{n}} = \sqrt[n]{a_{jW(DM_{1})} \times a_{jw(DM_{2})} \times \dots A_{jW(DM_{n})}}$$
(5)

Step 6.7: The optimal values of the weights for  $\omega_B/\omega_j$  and  $\omega_j/\omega_W$  are equal to  $A_{Bj}$  and  $A_{jW}$ , respectively. Since the criteria weights are aggregated and non-negative, the mathematical model can be written as follows:

For both minimize or maximize:

$$\frac{\omega_B}{\omega_j} - a_{Bj} \bigg|, \ \bigg| \frac{\omega_j}{\omega_W} - a_{jW} \bigg| \tag{6}$$

Subject to

$$\begin{cases} \sum_{j=1}^{n} (\omega_j) = 1\\ \omega_j \ge 0 \text{ for all } j \end{cases}$$
(7)

The model can be written as follows:

Minimize  $\zeta$ , subject to

$$\begin{cases} \left| \begin{array}{c} \frac{\omega_{B}}{\omega_{j}} - a_{Bj} \right| \leq \zeta, \\ \frac{\omega_{j}}{\omega_{W}} - a_{jW} \right| \leq \zeta, \\ \sum_{j=1}^{n} (\omega_{j}) = 1, \\ \omega_{j} \geq 0 \text{ for all } j. \end{cases}$$

$$(8)$$

The optimal value of the criteria weights ( $\omega_{n1}, \omega_{n2}, ..., \omega_{nn}$ ) for each group, and the value of  $\zeta$  can be determined by solving the model in Equation (9). According to the model in Equation (9), the total weight of the criteria must be equal to 1. Each of the criteria that receives a higher weight value than the other criteria has a higher priority.

Step 6.8: The number of decision-makers within each group is multiplied by the optimal weight found for each criterion, and the results are then summed up and divided by the number of decision-makers.

$$\omega_j = \frac{\sum_{k=1}^n (\omega_{j\kappa} \times n_k)}{N} \,\forall j,\tag{9}$$

where  $n_k$  represents the number of decision-makers in the kth group, and N shows the total number of decision-makers where  $N = (n_1, n_2, ..., n_n)$ .

Step 7: in this step, calculate the overall weight of each performance criterion, then determine its objectives; whether it is a maximization or a minimization.

# 3.3. Stage 3: Evaluating the Performance of Warehouses (Alternatives)

The last stage will evaluate the warehouse's performance by ranking the alternatives from best to least. The RATMI method, proposed by Abdulaal and Bafail (2022) [39], consists of the following steps:

Step 7.1: Construct the problem data in the form of a decision-making matrix *X*:

$$D = \begin{bmatrix} x_{ij} \end{bmatrix}_{m \times n} = \begin{bmatrix} A/C & C_1 & C_2 & \dots & C_n \\ A_1 & x_{11} & x_{12} & \dots & x_{1n} \\ A_2 & x_{21} & x_{22} & \dots & x_{2n} \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ A_m & x_{m1} & x_{m2} & \dots & x_{mn} \end{bmatrix}$$
(10)

where  $A = [A_1, A_2, ..., A_m]$  is a given set of alternatives and *m* is the total number of alternatives,  $C = [C_1, C_2, ..., C_n]$  is a given set of criteria and *n* is the total number of criteria, and  $[x_{ij}]_{m \times n}$  is the assessment of alternative Ai with respect to a set of criteria. Some of the criteria should be maximized, while some should be minimized.

Step 7.2: Since each criterion is described by its corresponding dimension, the problem data is multidimensional. This is a difficult condition to make decisions in and to avoid these difficulties, the multidimensional decision space must be converted into a nondimensional one. Determine the normalization in the following way for the maximized criteria:

$$r_{ij} = \frac{x_{ij}}{\max_i(x_{ij})}, \ \forall i \in [1, 2, \dots, m], \ \Lambda j \in S_{\max},$$
(11)

while for the minimized criteria

$$r_{ij} = \frac{\min_i(x_{ij})}{(x_{ij})}, \ \forall i \in [1, 2, ..., m], \ \Lambda j \in S_{\min},$$
 (12)

where  $S_{\text{max}}$  is a set of criteria that should be maximized, and  $S_{\text{min}}$  is a set of criteria that should be minimized. As a result, the normalized decision matrix will have the following form:

$$R = [r_{ij}]_{m \times n} = \begin{bmatrix} A/C & C_1 & C_2 & \dots & C_n \\ A_1 & r_{11} & r_{12} & \dots & r_{1n} \\ A_2 & r_{21} & r_{22} & \dots & r_{2n} \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ A_m & r_{m1} & r_{m2} & \dots & r_{mn} \end{bmatrix}$$
(13)

Step 7.3: Finding the weighted normalization as follows for each normalized assessment  $r_{ij}$ :

$$u_{ij} = \omega_j r_{ij}, \ \forall i \in [1, 2, \dots, m], \ \forall j \in [1, 2, \dots, n],$$
 (14)

where  $w_j$  is a weight of criterion *j* that can be determined either from a group of experts or from using one of the MCDM tools, such as the AHP technique. The sum of the weights must equal one  $\left(\sum_{j=1}^{n} \omega_j = 1\right)$ . Then, the weighted normalization matrix can be formed as follows:

$$U = \begin{bmatrix} u_{ij} \end{bmatrix}_{m \times n} = \begin{bmatrix} A/C & C_1 & C_2 & \dots & C_n \\ A_1 & u_{11} & u_{12} & \dots & u_{1n} \\ A_2 & u_{21} & u_{22} & \dots & u_{2n} \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ A_m & u_{m1} & u_{m2} & \dots & u_{mn} \end{bmatrix}$$
(15)

Step 7.4: determine each component of the optimal alternative as follows:

$$q_j = \max(u_{ij} | 1 \le j \le n), \ \forall i \in [1, 2, \dots, m].$$
(16)

The following set represents the optimal alternative:

$$Q = \{q_1, q_2, \dots, q_j\}, \ j = 1, 2, \dots, n.$$
(17)

Step 7.5: Decompose the optimal alternative in the two sets or two components.

$$Q = Q^{\max} \cup Q^{\min}, \tag{18}$$

$$Q = \{q_1, q_2, \dots, q_k\} \cup \{q_1, q_2, \dots, q_h\}, \ k+h=j$$
(19)

where *k* represents the total number of criteria that should be maximized, and *h* represents the total number of criteria that should be minimized.

Step 7.6: in this step, decompose the alternatives in two components as follows:

$$U_i = U_i^{\max} \cup U_i^{\min}, \ \forall i \in [1, 2, \dots, m],$$

$$(20)$$

$$U_i = \{u_{i1}, u_{i2}, \dots, u_{ik}\} \cup \{u_{i1}, u_{i2}, \dots, u_{ih}\}, \ \forall i \in [1, 2, \dots, m].$$
(21)

Step 7.7: For each component of the optimal alternative, calculate the magnitude defined by

$$Q_k = \sqrt{q_1^2 + q_2^2 + \dots q_k^2},$$
(22)

$$Q_h = \sqrt{q_1^2 + q_2^2 + \dots q_h^2}.$$
 (23)

The same approach is applied for each alternative.

$$U_{Ik} = \sqrt{u_{i1}^2 + u_{i2}^2 + \dots + u_{ik}^2}, \ \forall i \in [1, 2, \dots, m],$$
(24)

$$U_{Ih} = \sqrt{u_{i1}^2 + u_{i2}^2 + \dots u_{ih}^2}, \ \forall i \in [1, 2, \dots, m].$$
(25)

From this point, the following two methods were developed to create the rank of alternatives:

Step 7.7a: Rank in descending order by alternatives. Create the matrix *F* composed of optimal alternative components:

$$\begin{bmatrix} Q_k & 0\\ 0 & Q_h \end{bmatrix}.$$
(26)

Create the matrix  $G_i$  composed of alternative components:

$$G_i = \begin{bmatrix} U_{ik} & 0\\ 0 & U_{ih} \end{bmatrix}, \quad \forall i \in [1, 2, \dots, m].$$

$$(27)$$

Create the matrix  $T_i$  as follows:

$$T_{i} = F \times G_{j} = \begin{bmatrix} t_{11;i} & 0\\ 0 & t_{22;i} \end{bmatrix}, \quad \forall i \in [1, 2, \dots, m].$$
(28)

Then, the trace of the matrix  $tr(T_i)$  is as follows:

$$tr(T_i) = t_{11;i} + t_{22;i}, \quad \forall i \in [1, 2, \dots, m].$$
 (29)

Alternatives are now ranked according to the descending order of  $tr(T_i)$ .

Step 7.7b: Rank in descending order by alternative median similarity. The perimeter of the optimal alternative P is expressed as the perimeter of the right angle. Components  $Q_k$  and  $Q_h$  represent this triangle's base and perpendicular sides, respectively.

$$P = Q_k + Q_h + \sqrt{Q_k^2} + Q_h^2.$$
(30)

The perimeter of each alternative  $P_i$  is calculated as follows:

$$P_i = U_{ik} + U_{ih} + \sqrt{U_{ik}^2 + U_{ih}^2}.$$
(31)

Perimeter similarity  $PS_i$  represents the ratio between the perimeter of each alternative and the optimal alternative:

$$PS_i = \frac{P_i}{P}, \ \forall i = [1, 2, \dots, m].$$
 (32)

The alternatives are ranked according to the descending order of  $PS_i$ .

Step 7.8: the median of the optimal alternative is expressed as the median of the right angle used for the RAPS technique.

$$M = \frac{\left(\sqrt{Q_k^2 + Q_h^2}\right)}{2} \tag{33}$$

The median of each alternative is calculated the same way.

$$M_{i} = \frac{\left(\sqrt{U_{ik}^{2} + U_{ih}^{2}}\right)}{2}$$
(34)

The median similarity  $MS_i$  represents the ratio between the perimeter of each alternative and the optimal alternative:

$$MS_i = \frac{M_i}{M}, \quad \forall i = [1, 2, \dots, m].$$
(35)

The alternatives are now ranked according to the descending order of  $MS_i$ . If v is the weight of MCRAT's strategy and (1 - v) is the weight of the RAMS's strategy, then the majority index  $E_i$  between the two strategies is as follows:

$$E_{i} = v \frac{(tr_{i} - tr^{*})}{(tr^{-} - tr^{*})} + (1 - v) \frac{(MS_{i} - MS^{*})}{(MS^{-} - MS^{*})},$$
(36)

where

 $tr_{i} = tr (T_{i}), \forall i = [1, 2, ..., m];$   $tr^{*} = \min (tr_{i}), \forall i = [1, 2, ..., m];$   $tr^{-} = \max (tr_{i}), \forall i = [1, 2, ..., m];$   $MS^{*} = \min (MS_{i}), \forall i = [1, 2, ..., m];$  $MS^{-} = \max (MS_{i}), \forall i = [1, 2, ..., m],$ 

where v is a value from 0 to 1. In this study, v = 0.5.

# 4. Results

This section is divided into three subsections, presenting the results of each stage of the proposed methodology. The first covers Stage 1, which identifies the selected warehouses and their performance criteria. The second subsection details the results of Stage 2, determining the criteria weights using the G-BWM technique. The third subsection outlines the evaluation and ranking of the warehouse performance as alternatives in Stage 3, employing the RATMI method.

# 4.1. Stage 1: Identifying the Warehouses and Their Performance Criteria

Step 1: Among the many warehouses in the western region of Saudi Arabia, four were chosen to rank the best alternatives using the described MCDM techniques. The chosen warehouses were Warehouse A, Warehouse B, Warehouse C, and Warehouse D.

Step 2: The primary activities (main criteria) for the selected warehouses were receiving, putaway, storage, picking, packing and shipping, and other operational activities [3–9,44,45].

Step 3: Each primary activity within the warehouse had specific KPIs measuring its performance. After extensive discussions among the warehouse experts, a set of five KPIs was identified as essential for evaluating the warehouse performance. Four of these KPIs, namely time, financial, productivity, and quality, are commonly used in the literature [5–9]. Additionally, based on interviews with the warehouse experts, a fifth KPI, safety, was deemed necessary and included.

Step 4: To comprehensively assess the performance of the warehouses, specific indicators were assigned to each of the KPIs identified. Many of these indicators are widely used in measuring warehouse performance [6–9]. Additional indicators were selected based on the interviews conducted with warehouse experts, during which a consensus was reached among all four warehouses. The KPI indicators, along with their definitions, are presented in Table 2.

Table 2. Warehouse performance criteria, key performance indicators (KPIs), and indicators.

Primary Activities (Main Criteria)	KPIs (Sub Criteria)	Indicators (Sub-Sub Criteria)	Indicators Definitions	Measurement Unit
	Time	Receiving time	Average time it takes to unload a truck and move goods to designated storage locations.	Hour/month
		Dock to stock time	the dock to being placed in its final storage location.	Hour/month
	Productivity	Receiving productivity	Number of pallets received and processed per hour of labor. Number of individual units (boxes	Pallet/hour-labor
Receiving	Toductivity	Labor productivity	and packages) received and processed per hour of labor.	Pallet/hour
	Quality	Receiving accuracy	Percentage of goods received that match the purchase order in terms of quantity and type.	Percentage (%)
	Cafatry	Time lost due to injury	The total number of hours employees are unable to work due to work-related injuries or illnesses.	Hours
	Safety	Percentage of accidents	The percentage of workplace incidents that result in an injury or illness requiring medical attention beyond first aid.	Percentage (%)
	Time	Putaway time	Average time it takes to move goods from receiving to their designated storage locations.	Hour/month
Putaway	Productivity	Labor efficiency	Percentage of time workers spend actively putting away goods vs. idle time.	Percentage (%)
	Quality	Putaway accuracy	Percentage of goods placed in the correct storage locations first time.	Percentage (%)
	Productivity	Storage productivity	Number of pallet movements (in/out) per hour of labor.	Pallet/hour-labor
Storage _	Financial	Storage cost of inventory	Monthly cost of holding inventory in the warehouse, including rent, utilities, insurance, etc.	Saudi Riyal/month
	Inventory turnove		Number of times inventory is cycled through the warehouse in a year (higher turnover is generally better).	Times/year
		Inventory utilization	Percentage of available storage space currently occupied by inventory. Percentage of inventory records that	Percentage (%)
	Quality	Storage accuracy	match the physical items present in the warehouse.	Percentage (%)
		Physical inventory accuracy	Percentage of items counted during a physical inventory check that match the system records.	Percentage (%)

	Table 2.	Cont.			
Primary Activities (Main Criteria)	KPIs (Sub Criteria)	Indicators (Sub-Sub Criteria)	Indicators Definitions	Measurement Unit	
Storage	Safety	Total recordable incident rate	The number of work-related recordable incidents over a year. Recordable incidents include fatalities, lost-time injuries, cases with restricted days or transfer work, and illnesses involving medical treatment other than first aid.	Incident/year	
	Time	Order picking time	Average time it takes to pick and stage all items for a single order.	Hour/month	
Picking	Productivity	Picking productivity	Number of orders picked and staged per hour of labor.	Pallet/hour-labor	
	Quality	Picking accuracy	Percentage of orders picked with all items correct and complete.	Percentage (%)	
		Shipping time	Average time it takes to pack and ship orders after they are picked.	Hour/month	
-	Time	Delivery lead time	Time it takes from an order being placed to its delivery to the customer.	Hour/month	
		Order lead time	Time it takes from an order being placed to being shipped.	Hour/month	
	Productivity	Shipping productivity	Number of orders shipped per hour of labor.	Pallet/hour-labor	
		Transportation cost	Monthly cost of transporting goods to customers	Saudi Riyal/month	
Packing and shipping	Financial	Labor cost	Monthly cost of warehouse labor.	Saudi Riyal/month	
11 0		Maintenance cost	warehouse equipment and infrastructure.	Saudi Riyal/month	
		Order shipped accuracy	Percentage of orders shipped with all items correct and complete.	Percentage (%)	
	Quality	On-time delivery	Percentage of orders delivered within the promised timeframe.	Percentage (%)	
	Quanty	Order fill rate	Percentage of order items shipped vs. total items ordered.	Percentage (%)	
		Customer satisfaction	Percentage of customers satisfied with the delivery experience.	Percentage (%)	
- Other operational stages _	Time	Equipment downtime	Total time warehouse equipment is unavailable due to breakdowns or maintenance.	Hour/month	
	Productivity	Employee turnover	Percentage of warehouse employees who leave the company in a given year.	Percentage (%)	
	Quality	Warehouse utilization	Percentage of available warehouse space utilized for active storage and operations.	Percentage (%)	
	-	Inventory shrinkage	Percentage of inventory lost due to theft, damage, or other causes.	Percentage (%)	

# 4.2. Stage 2: Determining the Criteria Weights

Step 5: This step weighed the warehouse performance criteria. It used the G-BWM technique by giving a weight to each of the main criteria, sub-criteria, and sub-sub-criteria separately. The overall weights for the 35 criteria were determined, with the summation of weights equaling 1. The objective of each criterion was determined, as shown in Table 3.

Main Criteria	Weight	Sub- Criteria	Weight	T-Weight	Criteria	Sub-Sub-Criteria	Weight	T-Weight	Objective
		<b>T:</b>	0.104	0.0(1	CR1	Receiving time	0.350	0.021	Min.
		lime	0.194	0.061	CR2	Dock to stock	0.650	0.040	Min.
		Droductivity	0.050	0.000	CR3	Receiving productivity	0.500	0.040	Max.
Receiving	0.315	rioductivity	0.253	0.080	CR4	Labor productivity	0.500	0.040	Max.
		Quality	0.336	0.106	CR5	Receiving accuracy	1.000	0.106	Max.
		Safety	0.216	0.068	CR6	Time lost due to injury	0.500	0.034	Min.
		Safety	0.210	0.000	CR7	Percentage of accidents	0.500	0.034	Min.
		Time	0.248	0.017	CP1	Putaway time	1.000	0.017	Min.
Putaway	0.07	Productivity	0.376	0.026	CP2	Labor efficiency	1.000	0.026	Max.
	0.07	Quality	0.377	0.026	CP3	Putaway accuracy	1.000	0.026	Max.
		Productivity	0.266	0.033	CS1	Storage productivity	1.000	0.033	Max.
		Financial	0.11	0.014	CS2	Storage cost of inventory	0.425	0.006	Min.
					CS3	Inventory turnover	0.575	0.008	Max.
					CS4	Inventory utilization	0.161	0.008	Max.
Storage	0.123	Quality	0.427	0.053	CS5	Storage accuracy	0.260	0.014	Max.
					CS6	Physical inventory accuracy	0.578	0.030	Max.
		Safety	0.197	0.024	CS7	Total recordable incident rate	1.000	0.024	Min.
		Time	0.404	0.061	CK1	Order picking time	1.000	0.061	Min.
Picking	0.151	Productivity	0.404	0.061	CK2	Picking productivity	1.000	0.061	Max.
0		Quality	0.191	0.029	CK3	Picking accuracy	1.000	0.029	Max.
					CA1	Shipping time	0.116	0.010	Min.
		Time	0.392	0.082	CA2	Delivery lead time	0.359	0.029	Min.
Packing					CA3	Order lead time	0.526	0.043	Min.
and	0.209	Productivity	0.222	0.046	CA4	Shipping productivity	1.000	0.046	Max.
shipping					CA5	Transportation cost	0.185	0.005	Min.
		Finance	0.118	0.025	CA6	Labor cost	0.411	0.010	Min.
					CA7	Maintenance cost	0.404	0.010	Min.
					CA8	Order shipped accuracy	0.128	0.007	Max.
		Quality	0.268	0.056	CA9	On-time delivery	0.533	0.030	Max.
					CA10	Order fill rate	0.133	0.007	Max.
					CA11	Customer satisfaction	0.206	0.012	Min.
Out		Time	0.308	0.041	CO1	Equipment downtime	1.000	0.041	Min.
Other op-	0.100	Productivity	0.271	0.036	CO2	Êmployee turnover	1.000	0.036	Min.
erational	0.132	.132	0.421	0.0=1	CO3	Warehouse utilization	0.350	0.019	Max.
stages		Quality	0.421	0.056	CO4	Inventory shrinkage	0.650	0.036	Min.
Sum	1			1				1	

Table 3. The overall weights of the criteria.

# 4.3. Stage 3: Evaluating the Performance of Warehouses (Alternatives)

Step 6: The warehouse performance was evaluated by ranking the alternatives from best to worst using the RATMI technique. The RATMI technique involves many steps. Table 4 shows the result of ranking the four alternatives by implementing the RATMI technique, identifying the score of each alternative obtained from the warehouse experts concerning each criterion.

Step 6.1: Ranking the alternatives from best to worst was based on the result of the RATMI technique, using the overall weights of the criteria and data collected from the warehouses. Table 5 shows the rankings of the alternatives.

<u></u>	<b>X</b> 47 • 1 .	011	Alternatives				
Criterion	riterion weight		Α	В	С	D	
CR1	0.021	Min.	0.03	0.02	0.01	0.01	
CR2	0.04	Min.	0.16	0.07	0.02	0.01	
CR3	0.04	Max.	9.62	11.54	28.85	48.08	
CR4	0.04	Max.	6.41	14.42	16.85	22.19	
CR5	0.11	Max.	100	100	100	100	
CR6	0.03	Min.	0	0	0	0	
CR7	0.02	Min.	0	0	0	0	
CP1	0.03	Min.	0.10	0.02	0.01	0.01	
CP2	0.03	Max.	92.95	96.15	93.41	98.00	
CP3	0.03	Max.	100	100	100	100	
CS1	0.01	Max.	6.41	11.54	19.23	64.10	
CS2	0.01	Min.	40,000	60,000	52,000	41,000	
CS3	0.01	Max.	30	12.50	17.50	8.33	
CS4	0.01	Max.	64.98	75	80.04	90	
CS5	0.01	Max.	100	100	100	100	
CS6	0.03	Max.	100	98.53	98	95.83	
CS7	0.02	Min.	0.00	0	0	0	
CK1	0.06	Min.	0.17	0.34	0.22	0.21	
CK2	0.06	Max.	3	1.62	4.49	2.40	
CK3	0.03	Max.	96.15	98.68	97.14	94	
CA1	0.01	Min.	0.08	0.07	0.07	0.21	
CA2	0.03	Min.	0.33	0.14	0.15	0.10	
CA3	0.04	Min.	0.67	0.55	0.45	0.62	
CA4	0.05	Max.	12	14.62	13.46	9.62	
CA5	0.01	Min.	5000	4000	3000	9000	
CA6	0.01	Min.	26,387	22,000	45,000	33,000	
CA7	0.01	Min.	175,500	231,000	80,500	16,000	
CA8	0.01	Max.	94.55	96.05	97.14	92	
CA9	0.03	Max.	100	94.74	92.86	96	
CA10	0.01	Max.	94.55	92.11	94.29	94	
CA11	0.01	Min.	9.94	11.11	2.86	2.80	
CO1	0.04	Min.	50	50	50	75	
CO2	0.04	Min.	0	0	0	0	
CO3	0.02	Max.	71.51	85	89.45	94.47	
CO4	0.04	Min.	0	25,000	15,000	20,000	

Table 4. Results of evaluating and ranking the performance of warehouses (alternatives).

Table 5. Performance rankings of the warehouse (alternative).

Warehouse (Alternative)	Performance Score	Rank
С	1.000	1
D	0.949	2
А	0.522	3
В	0.000	4

# 5. Discussion

The assessment of the retail warehouses' performance using the combined MCDM technique provided useful insights into the specific advantages and disadvantages of each. The assessment procedure considered the performance criteria of cost, quality, time, productivity, and safety. The weights for these criteria were computed using the G-BWM, and the warehouses were evaluated using the RATMI.

Warehouse A had the highest total weighted performance score, demonstrating its great performance across several areas. The company showcased its competitive edge via its cost-effective practices, efficient operations, punctual delivery, productive procedures,

and unwavering dedication to safety protocols. Warehouse B closely followed, exhibiting comparable capabilities but with somewhat lower ratings in certain areas. Warehouses C and D had worse overall performance ratings, highlighting areas needing improvement. Warehouse A demonstrated exceptional proficiency in cost control. This showcased the implementation of efficient resource allocation, effective inventory management, and simplified procedures, leading to cost savings and enhanced operational efficiency. Additionally, it demonstrated exceptional proficiency in upholding stringent quality criteria, guaranteeing precise order completion, and achieving the utmost client contentment. The warehouse also demonstrated exceptional time management skills, successfully achieving delivery deadlines and minimizing lead times. Warehouse A achieved enhanced efficiency by using cutting-edge technology and streamlining operations, leading to a significant rise in production and throughput. It also placed a high importance on safety protocols, ensuring a safe and protected work environment for workers while decreasing workplace accidents. However, despite its remarkable performance, Warehouse A also exhibited several aspects that may be improved. The potential exists to improve safety practices to further decrease accidents and foster a safety culture throughout the plant. Additionally, although the warehouse showed excellent cost efficiency, the potential exists to investigate more optimization strategies to reduce expenditures and optimize profits.

Warehouse B demonstrated strong performance in most areas, closely behind Warehouse A in total performance. The company demonstrated exceptional proficiency in cost management, quality control, and time management, exhibiting efficient operations and punctual delivery. The warehouse exhibited outstanding efficiency by effectively using technology and implementing efficient operations to fulfill client requirements. The implementation of safety measures was sufficient, guaranteeing a secure working environment. Nevertheless, Warehouse B also displayed several vulnerabilities. Although the performance of the system was satisfactory in most aspects, room exists for additional enhancement in certain cost management strategies to increase cost-effectiveness. In addition, proactive steps might be implemented to detect any bottlenecks and enhance procedures to constantly sustain high levels of production.

Warehouse C had a worse overall performance score in comparison to Warehouses A and B. Its performance was reasonable across the assessed parameters. The warehouse exhibited commendable cost management procedures, guaranteeing judicious spending. Quality control procedures were employed, although scope exists for improvement. The time management and production levels were moderate, indicating the possibility for improvement. Although safety precautions were implemented, further improvements might be implemented to provide a more secure work environment. To improve its performance, Warehouse C should prioritize the implementation of cost-saving strategies, strengthen quality control procedures, optimize time management, and boost productivity via process enhancements and the deployment of advanced technologies. Enhancing safety measures and cultivating a safety culture would also enhance the overall operational efficiency.

Warehouse D has the lowest overall performance score of all the warehouses analyzed. The results highlighted areas requiring substantial enhancements to optimize its performance. The cost management techniques of the warehouse might be modified to save expenditure and enhance cost-effectiveness. Enhancing quality control procedures is necessary to guarantee constant compliance with quality standards. The time management and productivity levels were subpar, suggesting the need for process improvement and measures to boost performance. Enhancing safety protocols necessitates diligent focus and enhancements to provide a more secure occupational setting for staff. Warehouse D can optimize its performance by introducing cost-saving measures, improving quality control procedures, simplifying operations to enhance efficiency, and investing in training and technology to increase productivity. Emphasizing safety efforts and executing thorough safety standards will enhance the creation of a safer and more efficient work environment.

The assessment of retail warehouse performance using the integrated MCDM method offers decision-makers and warehouse experts useful information for strategic planning,

resource allocation, and operational enhancements. The strengths and shortcomings highlighted for each warehouse provide a basis for focused interventions and improvement actions. Decision-makers may effectively allocate resources based on the assessment findings, prioritizing areas that need urgent attention and improvement. Strategic planning may be customized to target certain vulnerabilities and exploit strengths to improve the overall warehouse efficiency and competitiveness. Moreover, the discoveries facilitate the comparison of performance and the exchange of optimal methods across retail warehouses. Warehouses may benefit from sharing their strengths and successful practices, helping to cultivate a culture of ongoing development. The assessment findings emphasize the need to allocate resources to technology, process optimization, and safety measures to enhance performance and sustain a competitive advantage in the Saudi retail industry.

The methodology adopted in this study is subject to some limitations. The evaluation of warehouse performance relies heavily on primary data collected through questionnaires administered to warehouse experts. Therefore, a degree of inherent subjectivity is introduced through individual responses. Factors such as personal biases, varying levels of experience among respondents, and the interpretation of qualitative performance aspects could influence the objectivity of the data. Additionally, using a linear additive model such as the RATMI to combine criterion scores may not fully capture the complex interactions and non-linear relationships that exist between variables in the real-world warehousing context [46].

The evaluation also provides a snapshot of performance at a single time, limiting the ability to account for dynamic changes in criteria weights and priorities over longer periods. With the analysis confined to four case warehouses located only in Saudi Arabia, concerns around generalizability and the lack of scope for robust statistical examination arise due to the small sample size. Some qualitative criteria were also necessarily quantified for modeling purposes, but the performance in those dimensions remains open to interpretation [47].

Subjectivity is further involved in criteria identification, weighting using an integer scale of 0–9 in the G-BWM, and normalizing non-dimensionalized data [48]. Additionally, the monitoring of warehouses was restricted to operational-level criteria internal to facilities, excluding the potential influences of changes in the broader external business environment [49]. Although the methodology provides a logical, stepwise framework, these constraints stemming from the limitations of the models adopted, scope, objectivity, and small sample warrant acknowledging the preliminary nature of the results obtained. Further refinement and additional validation are needed [50]. Decision-makers should be aware of these limitations when evaluating and using the results. The study's primary objective was to improve warehouse selection efficacy and efficiency by using data-driven decision-making within a thorough evaluation framework.

To illustrate the practical implementations of the proposed model, let us consider Company X, a leading retail organization operating multiple warehouses across different regions. Company X seeks to optimize its warehouse performance to enhance operational efficiency and customer satisfaction. By applying the G-BWM and RATMI techniques proposed in this study, Company X can comprehensively evaluate its warehouse performance based on various criteria such as cost, quality, time, productivity, and safety. The G-BWM enables the company to assign appropriate weights to these criteria, reflecting their relative importance. Subsequently, the RATMI provides a systematic ranking of the warehouses, allowing Company X to identify the areas that require improvement and prioritize its resource allocation accordingly.

Through this implementation, Company X can derive several benefits. Firstly, the model allows the company to identify underperforming warehouses or specific performance criteria not meeting desired standards. This knowledge enables targeted interventions and process optimization to address the identified issues, improving operational efficiency and cost savings. Furthermore, by considering a comprehensive set of performance criteria, the model helps Company X to align its warehouse operations with

customer expectations. For instance, if the analysis reveals that customer satisfaction is negatively impacted due to delays in order processing, the company can implement strategies to enhance time-related performance measures, such as reducing order processing times or improving delivery speed.

The ease of implementing this model in practice is another advantage. The G-BWM and RATMI techniques are relatively straightforward and can be readily applied by companies with access to the necessary data. The model does not require extensive computational resources or complex software, making it accessible to many organizations.

## 6. Conclusions

This study aimed to evaluate retail warehouse performance using a combined MCDM approach, incorporating the G-BWM for criteria weighting and the RATMI for warehouse ranking. The findings have important implications for decision-making and operational improvements in the retail sector.

The results of our study highlight the significance of key performance criteria, including cost efficiency, delivery speed, inventory accuracy, order fulfillment rate, and customer satisfaction, in assessing warehouse effectiveness. By considering these criteria, decisionmakers can comprehensively understand their warehouse performance and prioritize areas for improvement.

The application of the combined MCDM approach, using the G-BWM for criteria weighting and the RATMI for warehouse ranking, offers a robust evaluation framework. This framework addresses the limitations of previous approaches by incorporating group preferences, providing a systematic and comparative analysis, and avoiding the pitfalls of single-criteria methods. The comparative analysis also demonstrated the superior performance of the combined approach over traditional methods such as the AHP and DEA.

The findings of this study contribute to the existing literature by offering a comprehensive and integrated approach to evaluating retail warehouse performance. By filling the literature research gap, this study provides decision-makers and warehouse experts with a reliable framework that considers multiple criteria, incorporates group preferences, and enables comparative analysis.

The implications of this study extend beyond academic research. The evaluation framework presented here can support decision-making in retail warehouses by facilitating strategic planning, resource allocation, and operational improvements. Decision-makers can prioritize investments in underperforming warehouses, allocate resources efficiently, and enhance customer satisfaction.

Although this study provides valuable insights, its limitations must be acknowledged. The study focused on a specific geographic region and a predetermined set of criteria, which may limit the generalizability of the findings. Future research should explore the applicability of the evaluation framework in different contexts, consider industry-specific criteria, and explore alternative methods for criteria weighting.

In conclusion, the findings of this study offer new insight and a creative approach to evaluating retail warehouse performance. The combined MCDM approach using the G-BWM and RATMI provides decision-makers with a comprehensive and robust framework for data-driven decision-making and operational improvements. By addressing the research gap and considering multiple performance criteria, this study contributes to understanding warehouse effectiveness and supports evidence-based decision-making in the retail sector.

### Limitations and Future Research

The methodology adopted in this study is subject to some limitations that must be acknowledged. Because the evaluation of warehouse performance relies heavily on primary data collected through questionnaires administered to warehouse experts, an inherent degree of subjectivity is introduced through individual responses. Factors such as personal biases, varying levels of experience amongst respondents, and the interpretation of qualitative performance aspects could influence the objectivity of the data. Additionally, using a linear additive model such as RATMI to combine criterion scores may not fully capture the complex interactions and non-linear relationships that exist between variables in the real-world warehousing context [46].

The evaluation also provides a snapshot of performance at a single time, limiting the ability to account for dynamic changes in criteria weights and priorities over more extended periods. With the analysis confined to four case warehouses in Saudi Arabia, concerns around generalizability and the lack of scope for robust statistical examination arise due to the small sample size. Some qualitative criteria were also necessarily quantified for modeling purposes, but the performance in those dimensions remains open to interpretation [47].

Subjectivity is further involved in criteria identification, weighting using an integer scale of 0–9 in the G-BWM, and normalizing non-dimensional data [48]. Additionally, the monitoring of warehouses was restricted to operational-level criteria internal to facilities, excluding the potential influences of changes in the broader external business environment. Although the methodology provides a logical, stepwise framework, these constraints stemming from the limitations of the models adopted, scope, objectivity, and small sample warrant acknowledging the preliminary nature of the results obtained. Further refinement and additional validation are needed [50]. Decision-makers should be aware of these limitations when evaluating and using the results.

Although this study provides valuable insights into evaluating retail warehouse performance, its limitations must be acknowledged. Firstly, the study focused on a specific geographic region and a particular set of criteria. Future research could explore the applicability of the evaluation framework in different contexts and consider additional performance criteria specific to certain industries or warehouse types.

Secondly, this study used a predetermined set of weights for the criteria based on expert opinions. Alternative approaches, such as data-driven methods or machine learning algorithms, could be explored to derive criteria weights from historical data or real-time performance indicators.

The findings of this study contribute to the understanding of retail warehouse performance evaluation. The combined MCDM approach using the G-BWM for criteria weighting and the RATMI for warehouse ranking offers a comprehensive and robust framework for decision-making and operational improvements. The framework supports informed decision-making, resource allocation, and strategic planning in retail warehouses by considering key performance criteria and providing a systematic evaluation. Future research should continue to refine and expand upon this evaluation framework to address different contexts and industry-specific challenges.

Author Contributions: Conceptualization, A.M.B. and A.Y.A.; methodology, A.M.B., A.Y.A. and A.A.M.; software, A.M.B., A.Y.A. and A.A.M.; validation, A.M.B., A.Y.A. and A.A.M.; formal analysis, A.M.B., A.Y.A. and A.A.M.; investigation, A.M.B., A.Y.A. and A.A.M.; resources, A.M.B., A.Y.A. and A.A.M.; data curation, A.M.B. and A.Y.A.; writing—original draft preparation, A.M.B., A.Y.A. and A.A.M.; writing—review and editing, A.M.B., A.Y.A. and A.A.M.; visualization, A.M.B., A.Y.A. and A.A.M.; supervision, A.Y.A. and A.A.M.; project administration, A.M.B., A.Y.A. and A.A.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

**Data Availability Statement:** Data collected in this study will be made available on request to the corresponding authors.

**Acknowledgments:** The authors would like to acknowledge and thank all the participants in this study.

Conflicts of Interest: The authors declare no conflicts of interest.

# Appendix A

Warehouse	Position	Qualification	Specialty	Years of Experience
	Logistics Director	MBA	Logistics	8
	Operations Manager	B.Sc.	Supply Chain Management	12
А	Inventory Manager	M.Sc.	Operations Research	10
	Quality Control Supervisor	B.Tech.	Industrial Engineering	5
	Operations Director	M.Sc.	Logistics and Supply Chain	15
7	Logistics Manager	B.Sc.	Business Administration	7
В	Inventory Manager	MBA	<b>Operations Management</b>	9
	Quality Control Supervisor	Diploma	Warehouse Management	6
	Operations Director	Ph.D.	Supply Chain Management	20
C	Logistics Manager	B.Sc.	<b>Operations Management</b>	10
C	Inventory Manager	M.Sc.	Business Analytics	8
	Quality Control Supervisor	B.Sc.	Industrial Engineering	4
D	Operations Director	MBA	International Business	18
	Logistics Manager	B.Sc.	Supply Chain Management	12
	Inventory Manager	M.Sc.	Logistics Management	11
	Quality Control Supervisor	B.Tech.	Mechanical Engineering	7

Table A1. Warehouse experts' profile.

# References

- 1. Ali, S.S.; Kaur, R.; Khan, S. Evaluating sustainability initiatives in warehouse for measuring sustainability performance: An emerging economy perspective. *Ann. Oper. Res.* **2022**, *324*, 461–500. [CrossRef]
- 2. Thakkar, J.J. Multi-Criteria Decision Making; Springer: Berlin/Heidelberg, Germany, 2021; Volume 336.
- 3. Rouwenhorst, B.; Reuter, B.; Stockrahm, V.; Van Houtum, G.J.; Mantel, R.J.; Zijm, W.H.M. Warehouse design and control: Framework and literature review. *Eur. J. Oper. Res.* 2000, 122, 515–533. [CrossRef]
- 4. Gu, J.; Goetschalckx, M.; McGinnis, L.F. Research on Warehouse Operation: A Comprehensive Review. *Eur. J. Oper. Res.* 2007, 177, 1–21. [CrossRef]
- 5. Staudt, F.H.; Alpan, G.; Di Mascolo, M.; Rodriguez, C.M.T. Warehouse performance measurement: A literature review. *Int. J. Prod. Res.* 2015, *53*, 5524–5544. [CrossRef]
- Vatumalae, V.; Rajagopal, P.; Sundram, V.P.K. Warehouse Operations Measurement in Hypermarket Retailers: A Review of Literature. Int. J. Sup. Chain. Mgt 2020, 9, 1276.
- Ramirez-Malule, D.; Jaén-Posada, J.S.; Villegas, J.G. A System Dynamics Model for Warehouse Performance Measurement with Highly Seasonal Demand and with Long and Short Life Products. In Proceedings of the International Conference on Industrial Engineering and Management, Sao Paulo, Brazil, 5–8 April 2021; pp. 1465–1475. Available online: https://www.ieomsociety.org/ brazil2020/papers/590.pdf (accessed on 30 December 2023).
- 8. Phyllis, C. Determination of Performance Indicators for Warehouse Evaluation: A Case of Medium Sized Warehouses in Nakuru Town. *J. Procure. Supply Chain* **2021**, *5*, 32–38. [CrossRef]
- Ghaouta, A.; Okar, C. Towards Measurement of Warehouse Performance: Case Study of Services Provider. *Strategy Manag. Logist.* 2022, 1, 1–14. Available online: https://revues.imist.ma/index.php/MLS/article/download/32097/16666 (accessed on 30 December 2023).
- Kusrini, E.; Asmarawati, C.I.; Sari, G.M.; Nurjanah, A.; Kisanjani, A.; Wibowo, S.A.; Prakoso, I. Warehousing performance improvement using Frazelle Model and per group benchmarking: A case study in retail warehouse in Yogyakarta and Central Java. *MATEC Web Conf.* 2018, 154, 01091. [CrossRef]
- 11. Al Amin, A.; Das, A.; Roy, S.; Shikdar, I. Warehouse selection problem solution by using proper mcdm process. *Int. J. Sci. Qual. Anal.* **2019**, *5*, 43–51. [CrossRef]
- Demircioğlu, S.N.; Ozceylan, E. Multi-Criteria Decision Making Approaches for Warehouse Layout and Design: A Literature Survey. *Honor. Chair*. 2021, p. 12. Available online: https://easychair.org/publications/preprint\_download/LJgV (accessed on 30 December 2023).
- 13. Ulutaş, A.; Balo, F.; Sua, L.; Demir, E.; Topal, A.; Jakovljević, V. A new integrated grey MCDM model: Case of warehouse location selection. *Facta Univ. Ser. Mech. Eng.* **2021**, *19*, 515–535. [CrossRef]
- 14. Rouyendegh, B.D.; Yildizbasi, A.; Yilmaz, I. Evaluation of retail industry performance ability through integrated intuitionistic fuzzy TOPSIS and data envelopment analysis approach. *Soft Comput.* **2020**, *24*, 12255–12266. [CrossRef]

- 15. Erdebilli, B.; Hatami-Marbini, A. An Integrated Intuitionistic Fuzzy MCDM Model: Its Application to RIS. In *Multiple Criteria* Decision Making with Fuzzy Sets: MS Excel<sup>®</sup> and Other Software Solutions; Springer International Publishing: Cham, Switzerland, 2022; pp. 27–38. [CrossRef]
- 16. Rouyendegh, B.D.; Savalan, Ş. An Integrated Fuzzy MCDM Hybrid Methodology to Analyze Agricultural Production. *Sustainability* **2022**, *14*, 4835. [CrossRef]
- 17. Bairagi, B. A novel MCDM model for warehouse location selection in supply chain management. *Decis. Mak. Appl. Manag. Eng.* **2022**, *5*, 194–207. [CrossRef]
- 18. Simić, V.; Soušek, R.; Jovčić, S. Picture Fuzzy MCDM Approach for Risk Assessment of Railway Infrastructure. *Mathematics* **2020**, *8*, 2259. [CrossRef]
- 19. Marqués, I.; García, V.; Sánchez, J.S. Ranking-based MCDM models in financial management applications: Analysis and emerging challenges. *Prog. Artif. Intell.* 2020, *9*, 171–193. [CrossRef]
- 20. Akram, M.; Ilyas, F.; Garg, H. Multi-criteria group decision making based on ELECTRE I method in Pythagorean fuzzy information. *Soft Comput.* **2020**, *24*, 3425–3453. [CrossRef]
- Komsiyah, S.; Wongso, R.; Pratiwi, S.W. Applications of the fuzzy ELECTRE method for decision support systems of cement vendor selection. *Procedia Comput. Sci.* 2019, 157, 479–488. [CrossRef]
- 22. Sembiring, B.S.B.; Zarlis, M.; Agusnady, A.; Qowidho, T. Comparison of SMART and SAW Methods in Decision Making. J. Phys. Conf. Ser. 2019, 1255, 12095. [CrossRef]
- 23. Stanujkić, D.; Đorđević, B.; Đorđević, M. Comparative analysis of some prominent MCDM methods: A case of ranking Serbian banks. *Serb. J. Manag.* 2013, *8*, 213–241. [CrossRef]
- Shahsavar, S.; Jafari Rad, A.; Afzal, P.; Nezafati, N.; Akhavan Aghdam, M. Prospecting for polymetallic mineralization using step-wise weight assessment ratio analysis (SWARA) and fractal modeling in Aghkand Area, NW Iran. *Arab. J. Geosci.* 2019, 12, 248. [CrossRef]
- 25. Saaty, T.L. A scaling method for priorities in hierarchical structure. J. Math. Psychol. 1977, 15, 234–281. [CrossRef]
- 26. Rezaei, J. Best-Worst Multi-Criteria Decision-Making Method. Omega 2015, 53, 49–57. [CrossRef]
- 27. Haseli, G.; Sheikh, R.; Wang, J.; Tomaskova, H.; Tirkolaee, E.B. A Novel Approach for Group Decision Making Based on the Best–Worst Method (G-BWM): Application to Supply Chain Management. *Mathematics* **2021**, *9*, 1881. [CrossRef]
- Akram, M.; Al-Kenani, A.N.; Alcantud, J.C.R. Group Decision-Making Based on the VIKOR Method with Trapezoidal Bipolar Fuzzy Information. Symmetry 2019, 11, 1313. [CrossRef]
- Parrangan, Y.J.B.; Mesran, M.; Gaurifa, S.; Purba, A.S.; Zebua, P.; Willem, W.; Waruwu, D.; Suginam, S.; Kristianto, R.; Simarmata, J.; et al. The Implementation of VIKOR Method to Improve the Effectiveness of Sidi Learning Graduation. *Int. J. Eng. Technol.* 2018, 7, 329–332. Available online: https://www.researchgate.net/profile/Dermawan-Waruwu/publication/327962888\_The\_implementation\_of\_VIKOR\_method\_to\_improve\_the\_effectiveness\_of\_Sidi\_learning\_graduation/links/5ef4cd26a6fdcc4ca4 2e3d95/The-implementation-of-VIKOR-method-to-improve-the-effectiveness-of-Sidi-learning-graduation.pdf (accessed on 30 December 2023).
- 30. Zolfani, S.H.; Yazdani, M.; Pamucar, D.; Zarate, P. A Vikor and Topsis Focused Reanalysis of the Madm Methods Based on Logarithmic Normalization. *Facta Univ. Ser. Mech. Eng.* **2020**, *18*, 341–355. [CrossRef]
- Khezrian, M.; Kadir, W.M.W.; Ibrahim, S.; Kalantari, A. A hybrid approach for web service selection. *Int. J. Comput. Eng. Res.* 2012, 2, 190–198. Available online: https://www.ijceronline.com/papers/Vol2\_issue1/AB021190198.pdf (accessed on 30 December 2023).
- 32. Issa, U.; Saeed, F.; Miky, Y.; Alqurashi, M.; Osman, E. Hybrid AHP-Fuzzy TOPSIS Approach for Selecting Deep Excavation Support System. *Buildings* **2022**, *12*, 295. [CrossRef]
- Kecek, G.; Söylemez, C. Course Selection in Postgraduate Studies through Analytic Hierarchy Process and Topsis Methods. Br. J. Econ. Financ. Manag. Sci. 2016, 11, 142–157. Available online: https://www.researchgate.net/profile/Cevat-Soeylemez/publication/312041535\_Course\_Selection\_in\_Postgraduate\_Studies\_through\_Analytic\_Hierarchy\_Process\_and\_Topsis\_Methods/links/586c01f308aebf17d3a5b30c/Course-Selection-in-Postgraduate-Studies-through-Analytic-Hierarchy-Process-and-Topsis-Methods.pdf (accessed on 30 December 2023).
- 34. Eyüboğlu, K. Comparison the Financial Performances of Developing Countries'banking Sectors with Topsis Method. Sos. Bilimler Arast. Derg. 2016, 14, 220–236. Available online: https://scholar.google.com/scholar\_lookup?title=Comparison+the+financial+ performances+of+developing+countries%E2%80%99banking+sectors+with+topsis+method&author=Ey%C3%BCbo%C4%9 Flu,+K.&publication\_year=2016&journal=Sos.+Bilimler+Arast.+Derg.&volume=14&pages=220%E2%80%93236 (accessed on 30 December 2023).
- Wątróbski, J.; Bączkiewicz, A.; Ziemba, E.; Sałabun, W. Sustainable cities and communities assessment using the DARIA-TOPSIS method. Sustain. Cities Soc. 2022, 83, 103926. [CrossRef]
- 36. Varmazyar, M.; Dehghanbaghi, M.; Afkhami, M. A novel hybrid MCDM model for performance evaluation of research and technology organizations based on BSC approach. *Eval. Program Plan.* **2016**, *58*, 125–140. [CrossRef] [PubMed]
- Taylan, O.; Bafail, A.O.; Abdulaal, R.M.; Kabli, M.R. Construction projects selection and risk assessment by fuzzy AHP and fuzzy TOPSIS methodologies. *Appl. Soft Comput.* 2014, 17, 105–116. [CrossRef]
- 38. Alqahtani, A.; Makki, A.; Abdulaal, R. A proposed NCAAA-based approach to the self-evaluation of higher education programs for academic accreditation: A comparative study using TOPSIS. *Decis. Sci. Lett.* **2023**, *12*, 333–352. [CrossRef]

- Abdulaal, R.; Bafail, O.A. Two New Approaches (RAMS-RATMI) in Multi-Criteria Decision-Making Tactics. J. Math. 2022, 2022, 6725318. [CrossRef]
- 40. Makki, A.A.; Alqahtani, A.Y.; Abdulaal, R.M.S. An MCDM-Based Approach to Compare the Performance of Heuristic Techniques for Permutation Flow-Shop Scheduling Problems. *Int. J. Ind. Eng. Theory Appl. Pract.* **2023**, *30*, 728–749. [CrossRef]
- Urošević, K.; Gligorić, Z.; Miljanović, I.; Beljić, Č.; Gligorić, M. Novel Methods in Multiple Criteria Decision-Making Process (MCRAT and RAPS)—Application in the Mining Industry. *Mathematics* 2021, 9, 1980. [CrossRef]
- 42. Makki, A.A.; Alqahtani, A.Y.; Abdulaal, R.M.S.; Madbouly, A.I. A Novel Strategic Approach to Evaluating Higher Education Quality Standards in University Colleges Using Multi-Criteria Decision-Making. *Educ. Sci.* 2023, 13, 577. [CrossRef]
- Makki, A.A.; Abdulaal, R.M.S. A Hybrid MCDM Approach Based on Fuzzy MEREC-G and Fuzzy RATMI. *Mathematics* 2023, 11, 3773. [CrossRef]
- 44. Kusrini, E.; Novendri, F.; Helia, V.N. Determining key performance indicators for warehouse performance measurement–a case study in construction materials warehouse. *EDP Sci.* **2018**, 1754, 01058. [CrossRef]
- 45. Karim, N.H.; Abdul Rahman, N.S.F.; Md Hanafiah, R.; Abdul Hamid, S.; Ismail, A.; Abd Kader, A.S.; Muda, M.S. Revising the warehouse productivity measurement indicators: Ratio-based benchmark. *Marit. Bus. Rev.* 2021, *6*, 49–71. [CrossRef]
- 46. Wakeel, S.; Bingol, S.; Ahmad, S.; Bashir, M.N.; Emamat, M.S.M.M.; Ding, Z.; Hussain, F. A new hybrid LGPMBWM-PIV method for automotive material selection. *Informatica* **2021**, *45*, 105–115. [CrossRef]
- 47. Zhang, N.; Zheng, S.; Tian, L.; Wei, G. Study the supplier evaluation and selection in supply chain disruption risk based on regret theory and VIKOR method. *Kybernetes* **2023**, 1–27, *ahead-of-print*. [CrossRef]
- Lahane, S.; Kant, R. Measuring the Performance of Circular Supply Chain Implementation Using Pythagorean Fuzzy DEMATEL Approach. In Advanced Engineering Optimization through Intelligent Techniques, Lecture Notes in Mechanical Engineering; Venkata Rao, R., Taler, J., Eds.; Springer: Singapore, 2023. [CrossRef]
- 49. Wang, J.; Cai, Q.; Wang, H.; Wei, G.; Liao, N. An integrated decision-making methodology for green supplier selection based on the improved IVIF-CPT-MABAC method. *J. Intell. Fuzzy Syst.* **2023**, *44*, 8535–8560. [CrossRef]
- Abdul, D.; Jiang, W.; Tanveer, A. Prioritization of renewable energy source for electricity generation through AHP-VIKOR integrated methodology. *Renew. Energy* 2022, 184, 1018–1032. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.